Author's response to reviews

Title: Validity and reproducibility of folate and vitamin B12 intakes estimated from a self-administered diet history questionnaire in Japanese pregnant women

Authors:

Mie Shiraishi (mi-shi@umin.ac.jp)
Megumi Haruna (mharuna-tky@umin.ac.jp)
Masayo Matsuzaki (msumi-tky@umin.ac.jp)
Ryoko Murayama (rymurayama-tky@umin.ac.jp)
Satoshi Sasaki (stssasak@m.u-tokyo.ac.jp)
Sachiyo Murashima (murasima-tky@umin.ac.jp)

Version: 2 Date: 31 January 2012

Author's response to reviews: see over
January 31st, 2012

Dr. Nehme Gabriel
Editor-in Chief, Nutrition Journal

Dear Dr. Nehme Gabriel,

Thank you for the review of our manuscript. We greatly appreciate the reviewer's helpful comments. We have revised the manuscript accordingly. The revised parts are highlighted in blue and underlining in the text.

Thank you for considering our manuscript for publication in Nutrition Journal. We look forward to hearing from you in the near future.

Sincerely yours,

Mie Shiraishi
Department of Midwifery and Women’s Health,
Graduate school of Health Science & Nursing,
The University of Tokyo
Tel / Fax : +81-3-5841-3396
Address : 7-3-1, Hongo, Bunkyo-ku,
          Tokyo, 113-0033, Japan
E-mail : mi-shi@umin.ac.jp